Author’s response to reviews

Title: The HIV/AIDS epidemic in Cuba: Why a low prevalence?

Authors:

Hector de Arazoza (arazoza@matcom.uh.cu)
Jose Joanes (ssida@informed.sld.cu)
Rachid Loumes (loumes@math-info.univ-paris5.fr)
Camille Legeai (camille.legeai@apr.aphp.fr)
Stephan Clemenceau (sclercnac@gmail.com)
Jorge Perez (jorge.perez@ipk.sld.cu)
Bertran Auvert (bertran.auvert@uvusq.fr)

Version: 3 Date: 25 October 2007

Author’s response to reviews:

One new author who has been forgotten: Camille Legeai

We have added one sentence at the end of the methods section: This study has been approved by the Ethics Committee of the Instituto de Medicina Tropical "Pedro Kouri", Havana Cuba in its monthly meeting of January 2005.

We have added a acknowledgement section: Acknowledgments

The study was funded by CNRS (AC-10), Paris, France, and the Institut National de la Santé et de la Recherche Médicale, Paris, France.